ORR Reference : [

]

Subject to Contract and Without Prejudice

DATED 9 January, 2015

Between

DB Cargo International Limited As Facility Owner

- and -

Direct Rail Services Limited As Beneficiary

FACILITY ACCESS CONTRACT - MOSSEND EUROTERMINAL



DB Cargo International Limited Steering & Processes c/o Down Side Offices Station Approach Westbury BA13 4HP

FADRS 16007 Sales Mossend v3-5b English

1

TABLE OF CONTENTS

CLA	USE NO:	PAGE NO:
1.	INTERPRETATION	1
1.1	Definitions	1
1.2	Interpretation	6
1.3	Schedules	7
2.	EFFECTIVE DATE AND DURATION	7
2.1	Effective Date	7
2.2	Duration	7
3.	STANDARD OF PERFORMANCE	7
3.1	General standard	7
3.2	Good faith	8
3.3	Compliance with laws	8
4.	PERMISSION TO USE THE FACILITY	8
4.1	Permission to use	8
4.2	Meaning	8
4.3	Permission under Clauses 4.2(e) and 4.2(f)	8
4.4	Stabling	9
5.	OPERATION AND MAINTENANCE	9
5.1	General	9
5.2	Safety	9
5.3	Movement of trains onto and off the facility	10
6.	DB Cargo's OBLIGATIONS	10
7.	DRS's OBLIGATIONS	11
8.	GOODS	11
8.1	Risk	11
8.2	Dangerous Goods	11
8.3	Ability to reject Dangerous Goods	12
9.	ACCESS CHARGES	12
9.1	Obligation on the DRS to pay	12
9.2	Price variation	12
9.3	Unpaid sums	13
10.	INDEMNITIES AND LIMITATION OF LIABILITY	14
10.1	DB Cargo indemnity	14
10.2	DRS indemnity	14
10.3	Mitigation	14
10.4	Notification of claims, limitation on liability and restriction on claims	14

11.	GOVERNING LAW	15
12.	DISPUTE RESOLUTION	16
13.	CONFIDENTIALITY	16
13.1	Confidential Information	16
13.2	Entitlement to divulge	16
13.3	Return of Confidential Information	17
13.4	Retention or destruction of Confidential Information	18
13.5	Ownership of Confidential Information	18
14.	ASSIGNMENT AND SUB-CONTRACTING	18
14.1	Assignment	18
14.2	Sub-contracting	18
15.	PAYMENTS, INTEREST AND VAT	18
15.1	Payment	18
15.2	Disputed amounts	19
15.3	Interest	19
15.4	VAT	19
16.	FORCE MAJEURE EVENTS	20
17.	MISCELLANEOUS	21
17.1	Non waiver	22
17.2	Variations	22
17.3	Entire contract and exclusive remedies	22
17.4	Notices	23
17.5	Counterparts	24
17.6	Survival	24
17.7	Contracts (Rights of Third Parties) Act 1999	25
SCH	EDULE 1 - CONTACT DETAILS	27
SCH	EDULE 2 – THE FACILITY, FACILITY OPERATING CONSTRAINTS, FA	CILITY SERVICES
AND	ANCILLARY SERVICES	29
Ар	pendix 1 – The Plan	31
SCH	EDULE 3 – THE TRAIN SERVICES	32
Ар	pendix 1 – Form of Specification	35
SCH	EDULE 4 – EVENTS OF DEFAULT, SUSPENSION AND TERMINATION	37
SCH	40	

THIS CONTRACT is made on the [G] day of [Joway] 20197

BETWEEN: .

- (1) DB Cargo International Limited a company registered in England and Wales under number 3232475, having its registered office at Lakeside Business Park, Carolina Way, Doncaster DN4 5PN ("DB Cargo"); and
- (2) Direct Rail Services Limited, a company registered in England and Wales under number 3020822, having its registered office at Herdus House, Westlakes Science and Technology Park, Moor Row, Cumbria CA24 3HU ("DRS").

WHEREAS: -

- (A) DB Cargo is the facility owner and operator of the facility.
- (B) DRS is a train operator that wishes to obtain permission to use the facility.
- (C) DB Cargo has agreed to grant DRS and its associates permission to use the facility on the terms and conditions of this contract.
- (D) This contract is entered into pursuant to directions given by the Office of Rail and Road in the exercise of its powers under section 18 of the Act.

IT IS AGREED as follows: -

1. INTERPRETATION

1.1 Definitions

In this contract, unless the context otherwise requires:

"Access Sum" is the amount set out in paragraph 1 of Schedule 5;

"Access Dispute Resolution Rules" means the set of rules entitled "Access Dispute Resolution Rules" annexed to the publication entitled "The Network Code";

"Act" means the Railways Act 1993;

"Adjustment Factor" has the meaning attributed to it in Clause 9.2;

"Affiliate" means in relation to any company:

- a) a company which is either a holding company or a subsidiary of such company; or
- b) a company which is a subsidiary of a holding company of which such company is also a subsidiary;

and for these purposes "holding company" and "subsidiary" have the meanings

ascribed to them in section 1159 of the Companies Act 2006;

"Ancillary Charges" are the amounts specified in paragraph 2 of Schedule 5 for provision of the Ancillary Services;

"Ancillary Movements" means train movements on the facility which are not an express part of the Train Services but which are necessary or reasonably required for giving full effect to the Train Services;

"Ancillary Services" are those services, which are not Facility Services, which may be provided by DB Cargo to DRS on request, as more particularly described in paragraph 7 of Schedule 2;

"associate" has the meaning ascribed to it in section 17 of the Act;

"Charges" means the total of the Access Sums and the Ancillary Charges payable each Charging Period under this contract;

"Charging Period" means a calendar month save that the first period and the last period may be of less than a calendar month if the Effective Date does not coincide with the first day of a calendar month or the Expiry Date does not coincide with the last day of a calendar month;

"Confidential Information" means information relating to the affairs of one party to this contract or any of its Affiliates which has been provided by any such person to the other party under or for the purposes of this contract, or any matter or thing contemplated by this contract or to which this contract relates, the disclosure of which is likely materially to compromise or otherwise prejudice the commercial interests of any such person;

"contract" means this document including all schedules and appendices to it;

"Contract Year" means a calendar year commencing at 0000 hours on 1 January and ending immediately before 0000 hours on the next succeeding 1 January save that:

- (a) the first such period shall commence on the Effective Date; and
- (b) the last such period shall end on the Expiry Date;

"Dangerous Goods" means any Goods incorporating dangerous substances listed in the Carriage of Dangerous Goods and Use of Transportable Pressure Equipment Regulations 2009 and any other substance of a similar nature or presenting a similar hazard;

"DB Cargo Event of Default" has the meaning attributed to it in paragraph 1.3 of Schedule 4;

"DRS Event of Default" has the meaning attributed to it in paragraph 1.1 of Schedule 4;

"Decision Criteria" means the necessity or desirability of the following:

- (a) sharing the capacity, and securing the development, of the facility and the Premises in the most efficient and economical manner in the interests of all users having regard, in particular, to safety, the environment, and the proper maintenance, improvement and usage of the facility and the Premises;
- (b) enabling users of the facility and the Premises to comply with any contracts to which they are party, in each case to the extent that DB Cargo has been informed of such contracts;
- (c) enabling DB Cargo to comply with any contracts to which it is a party; and
- (d) maintaining, renewing and carrying out other necessary work on or in relation to the facility or on the Premises;

"Default Interest Rate" is two percent above the base lending rate of Barclays Bank PLC, as varied from time to time;

"Dispute" has the meaning attributed to it in Clause 12.1;

"Disruptive Event" means any event or circumstance which materially prevents or materially disrupts the operation of trains on any relevant part of the facility, but which is not a Force Majeure Event;

"Effective Date" means the date of signature of this contract;

"Environmental Damage" means any injury or damage to persons, living organisms or property (including offence to man's senses) or any pollution or impairment of the environment resulting from the discharge, emission, escape or migration of any substance, energy, noise or vibration;

"Event of Default" means a DRS Event of Default or a DB Cargo Event of Default as the context requires;

"Expiry Date" means the date on which this contract expires in accordance with Clause 2.2;

"facility" means the railway network on the Premises of which DB Cargo is the facility owner, and which is more particularly described in Schedule 2 and depicted on the Plan;

"Facility Operating Constraints" means the physical and operational constraints of the facility to accommodate the operation of trains as specified in paragraph 5 of Schedule 2 or as published from time to time by DB Cargo;

"facility owner" has the meaning ascribed to it under section 17(6) of the Act;

"Facility Services" are those services to be provided by DB Cargo to DRS and which are included in the Access Sum, as more particularly described in paragraph 6 of Schedule 2;

"Facility Workplan" means the plan drawn up by DB Cargo in respect of train

movements on the facility and provided to DRS in accordance with Clause 6.3;

"Force Majeure Event" has the meaning attributed to it in Clause 16;

"Goods" means cargo of any description whatsoever, together with any package, case, pallet, container or other thing which contains, protects or supports cargo or is designed or made to do so;

"Insolvency Event" means, in relation to either of the parties, an event which has occurred where:

- (a) any step which has a reasonable prospect of success is taken by any person with a view to its administration under Part II of the Insolvency Act 1986;
- (b) it stops or suspends or threatens to stop or suspend payment of all or a material part of its debts, or is unable to pay its debts, or is deemed unable to pay its debts under Section 123(1) or (2) of the Insolvency Act 1986, except that in the interpretation of this paragraph:
 - (i) Section 123(1)(a) of the Insolvency Act 1986 shall have effect as if for "£750" there were substituted "£100,000" or such higher figure as the parties may agree in writing from time to time; and
 - (ii) it shall not be deemed to be unable to pay its debts for the purposes of this paragraph if any such demand as is mentioned in Section 123(1)(a) of the Insolvency Act 1986 is satisfied before the expiration of 21 days from such demand;
- (c) its directors make any proposal under Section 1 of the Insolvency Act 1986, or it makes any contract for the deferral, rescheduling or other readjustment (or makes a general assignment or an arrangement or composition with or for the benefit of the relevant creditors) of all or a material part of its debts, or a moratorium is agreed or declared in respect of or affecting all or a material part of its debts;
- (d) any step is taken to enforce security over or a distress, execution or other similar process is levied or sued out against the whole or a substantial part of its assets or undertaking, including the appointment of a receiver, administrative receiver, manager or similar person to enforce that security;
- (e) any step is taken by any person with a view to its winding-up or any person presents a winding-up petition which is not dismissed within 14 days, or it ceases or threatens to cease to carry on all or a material part of its business, except for the purpose of and followed by a reconstruction, amalgamation, reorganisation, merger or consolidation on terms approved by the other party before that step is taken (which approval shall not be unreasonably withheld or delayed); or
- (f) any event occurs which, under the law of any relevant jurisdiction, has an analogous or equivalent effect to any of the events listed above,

unless:

- (i) in any case, a railway administration order (or application for such order) has been made or such order (or application) is made within 14 days after the occurrence of such step, event, proposal or action (as the case may be) in relation to that party pursuant to Sections 60, 61 or 62 of the Act and for so long as any such order (or application) remains in force or pending; or
- (ii) in the case of paragraphs (a), (d), and (e), the relevant petition, proceeding or other step is being actively contested in good faith by that party with timely recourse to all appropriate measures and procedures;

"Liability Cap" means the sum calculated in accordance with Clause 10.4.2;

"Network" means the network in respect of which Network Rail is the facility owner and which is situated in England, Wales and Scotland;

"Network Rail" means Network Rail Infrastructure Limited, incorporated in England and Wales under registered number 2904587;

"Office of Rail and Road" has the meaning ascribed to it under Section 15 of the Railways and Transport Safety Act 2003, and **"ORR"** shall be construed accordingly;

"Operational Stabling" means any parking or laying up of Specified Equipment which is included in the Specification and which is required to give effect to the Train Services;

"Plan" means the plan of the Premises, including the facility, appended to Schedule 2;

"**Premises**" means the rail terminal and associated facilities operated by DB Cargo and situated adjacent to the Mossend Eurocentral complex near Motherwell in the county of Lanark;

"**RPI**" has the meaning attributed to it in Clause 9.2.7;

"Safe System of Work" means an agreed statement of the procedures to be followed by DB Cargo and DRS in respect of the safe operation of the facility and trains thereon;

"Safety Obligations" means all applicable obligations concerning health and safety (including any duty of care arising at common law, and any obligation arising under statute, statutory instrument or mandatory code of practice) in Great Britain;

"Special Instruction" means an instruction issued by DB Cargo to DRS in respect of any short term operational or safety requirement not covered by a Safe System of Work or a Standing Instruction;

"Specification" means information provided to DB Cargo by DRS regarding the

Train Services in accordance with Clause 7.1 and in the form appended to Schedule 3;

"Specified Equipment" means railway vehicles, including Traction Units, registered for operation on the Network, and which DRS is therefore permitted to use on the track comprised in the facility, as more particularly described in Schedule 3;

"Stabling" means any parking or laying up of Specified Equipment between the time of its delivery to the facility and the time of its collection from the facility which is not Operational Stabling, together with any marshalling, and "Stabled" and "Stable" shall be construed accordingly;

"Standing Instruction" means an instruction issued by DB Cargo to DRS to cover any permanent operational or safety requirement not covered by a Safe System of Work;

"Suspension Notice" means a notice served by one party on the other pursuant to paragraph 2 of Schedule 4;

"Terminal" means a terminal on the Premises at which Goods are loaded onto or unloaded from railway vehicles;

"Termination Notice" means a notice served by one party on the other pursuant to paragraph 3 of Schedule 4;

"Traction Units" means a diesel locomotive, diesel multiple-unit or other railway vehicle of a type listed in paragraph 2 of Schedule 3;

"Train Services" means those train services operated by DRS and included in the Specification;

"Working Day" means each of Monday to Friday (inclusive) excluding common law and statutory public holiday; and

"year" means, for the purpose of Clause 2.2, a period of 12 months commencing on the Effective Date and on each successive anniversary of the Effective Date and ending on the day before each successive anniversary of the Effective Date.

1.2 Interpretation

In this contract, unless the context otherwise requires:

- (a) the singular includes the plural and vice versa;
- (b) any one gender includes the other;
- (c) all headings are for ease of reference only and shall not be used in the construction of this contract;
- (d) reference to an item of primary or secondary legislation is to that item as amended or replaced from time to time;

- (e) reference to a contract, instrument or other document is to that contract, instrument or other document as amended, novated, supplemented or replaced from time to time;
- (f) reference to a party is to a party to this contract, its successors and permitted assigns;
- (g) reference to a Clause or Schedule is to a clause or schedule of or to this contract; reference in a schedule to a Part of or an Appendix to a schedule is to a part of or an appendix to the schedule in which the reference appears; and reference in a Part of a Schedule to a paragraph is to a paragraph of that part;
- (h) where a word or expression is defined, cognate words and expressions shall be construed accordingly;
- (i) references to the word "person" or "persons" or to words importing persons include individuals, firms, corporations, government agencies, committees, departments, authorities and other bodies incorporated or unincorporated, whether having separate legal personality or not;
- (j) "otherwise" and words following "other" shall not be limited by any foregoing words where a wider construction is possible;
- (k) the words "including" and "in particular" shall be construed as being by way of illustration or emphasis and shall not limit or prejudice the generality of any foregoing words; and
- (1) words and expressions defined in the Act shall, unless otherwise defined in this contract, have the same meanings in this contract.

1.3 Schedules

The Schedules to this contract shall have effect.

2. EFFECTIVE DATE AND DURATION

2.1 Effective Date

The provisions of this contract take effect from the Effective Date.

2.2 Duration

Subject to the provisions for earlier termination in paragraph 3 of Schedule 4, this contract shall continue in force until 31st March 2019.

3. STANDARD OF PERFORMANCE

3.1 General standard

Without prejudice to all other obligations of the parties under this contract, each party shall, in its dealings with the other for the purpose of, and in the course of performance of its obligations under, this contract, act with due efficiency and

economy and in a timely manner with that degree of skill, diligence, prudence and foresight which should be exercised by a skilled and experienced:

- (a) facility owner and operator (in the case of DB Cargo); and
- (b) train operator (in the case of DRS).

3.2 Good faith

The parties to this contract shall, in exercising their respective rights and complying with their respective obligations under this contract (including when conducting any discussions or negotiations arising out of the application of any provisions of this contract or exercising any discretion under them), at all times act in good faith.

3.3 Compliance with laws

The parties to this contract shall at all times comply with all relevant legislation, regulations and bye-laws.

4. **PERMISSION TO USE THE FACILITY**

4.1 **Permission to use**

DB Cargo grants DRS permission to use the facility in accordance with the terms of this contract.

4.2 Meaning

References in this contract to permission to use the facility shall, except where the context otherwise requires, be construed to mean permission:

- (a) to use the track comprised in the facility for the operation of the Train Services using the Specified Equipment;
- (b) to make Ancillary Movements;
- (c) to Stable;
- (d) to load, unload and marshal the Train Services;
- (e) for DRS and its associates to enter upon and leave the facility with or without railway vehicles, including for the purpose of undertaking ad hoc repairs to the Specified Equipment; and
- (f) for DRS and its associates to bring such things onto the facility as are required in connection with the provision of the Train Services and keep them there,

and such permission is subject, in each case and in all respects, to the facility Operating Constraints.

4.3 Permission under Clauses 4.2(e) and 4.2(f)

4.3.1 Restrictions

In relation to the permissions specified in Clauses 4.2(e) and 4.2(f):

- (a) DRS shall, and shall procure that its associates shall, wherever reasonably practicable, first obtain the consent of DB Cargo, which consent shall not be unreasonably withheld or delayed;
- (b) DRS shall, and shall procure that its associates shall, remove any railway vehicle or other thing so brought onto any part of the facility when reasonably directed to do so by DB Cargo; and
- (c) DRS shall, and shall procure that its associates shall, whilst exercising any rights conferred by Clauses 4.2(e) and 4.2(f), comply with such reasonable restrictions or instructions as DB Cargo shall specify.

4.3.2 *Failure to comply with directions*

If DRS fails to comply with any directions given under Clause 4.3.1(b), DB Cargo shall be entitled to remove from the facility any railway vehicle or other thing or to instruct a third party to do so and any reasonable costs incurred by DB Cargo in taking such steps shall be paid promptly by DRS.

4.3.3 Evidence of costs

DB Cargo shall provide such evidence of such costs as are referred to in paragraph 4.3.2 as DRS shall reasonably request.

4.4 **Operational Stabling**

DB Cargo shall use its reasonable endeavours, having regard to the Facility Operating Constraints, the Decision Criteria and its Safety Obligations, to provide such Operational Stabling facilities as are necessary or expedient for or in connection with the operation of the Train Services and provision of the Facility Services.

5. **OPERATION AND MAINTENANCE**

5.1 General

Without prejudice to the other provisions of this contract:

- (a) DRS shall maintain and operate the Specified Equipment used on the facility in accordance with Clause 3.1 with a view to permitting the operation of the Train Services on the facility in accordance with the permission to use under this contract; and
- (b) DB Cargo shall maintain and operate the facility in accordance with Clause 3.1 with a view to permitting the operation of the Train Services on the facility using the Specified Equipment in accordance with the permission to use under this contract.

5.2 Safety

In relation to Safety Obligations:

(a) DRS shall comply with any reasonable request by DB Cargo in relation to any aspect of DRS's operations which affects or is likely to affect the performance of DB Cargo's Safety Obligations; and

(b) DB Cargo shall comply with any reasonable request by DRS in relation to any aspect of DB Cargo's operations which affects or is likely to affect the performance of DRS's Safety Obligations.

5.3 Movements of trains onto and off the facility

5.3.1 Suitable access

DB Cargo shall use its reasonable endeavours to ensure that the facility remains connected to the Network.

5.3.2 Prompt presentation

Where railway vehicles under the control of DRS will move onto and off the facility, the parties shall ensure that they facilitate (to the extent they are able) the prompt presentation of such railway vehicles onto and off the facility.

6. **DB CARGO'S OBLIGATIONS**

- 6.1 DB Cargo shall permit DRS to use the facility at such times and for such Train Services as, in each case, DRS may request in a Specification, except where and to the extent that DB Cargo reasonably determines, having regard to the Facility Operating Constraints, the Decision Criteria and its Safety Obligations, or as a result of a Disruptive Event, it is unable to do so.
- 6.2 DB Cargo shall, as soon as reasonably practicable, permit DRS to use the facility for any Train Service or train services not running in accordance with a Specification, except where, and to the extent that DB Cargo reasonably determines, having regard to the Facility Operating Constraints, the Decision Criteria and its Safety Obligations, or as a result of a Disruptive Event, it is unable to do so. Where DRS attempts to use the facility in respect of a train service which is not run in accordance with a Specification and for which insufficient or no advance notice has been given DB Cargo may:
 - (a) decline to permit DRS to use the facility for that train service; or
 - (b) charge DRS a Premium Rate, meaning the Access Sum specified in paragraphs 1.7 or 1.8 in Schedule 5, in respect of that train service and DRS rights under this Agreement will otherwise be unaffected.
- 6.3 DB Cargo shall provide the Facility Services in accordance with each agreed Specification. Where DRS provides a Specification pursuant to Clause 7.1 and DB Cargo is unable to provide the Facility Services in accordance with such Specification, DB Cargo shall use its reasonable endeavours to provide them at a time which is reasonably convenient for DRS, except where, and to the extent that, DB Cargo shall, having due regard to the Decision Criteria, and its Safety Obligations, or as a result of a Disruptive Event, reasonably determine that the work should not be carried out.
- 6.4 DB Cargo shall inform DRS if any Specified Equipment in respect of which Access is to be or is granted is, in DB Cargo's reasonable opinion, in an unsafe condition. DB

Cargo shall be under no obligation to provide Access, Facility Services, or Ancillary Services in respect of such Specified Equipment.

- 6.5 DB Cargo shall ensure that DRS is provided with a copy of the Facility Workplan and shall ensure that DRS is informed promptly, of any changes to it, if so requested.
- 6.6 DB Cargo shall give DRS at least 28 days' notice of planned possessions on the facility, and shall inform DRS of any Disruptive Event or emergency possessions on the facility as soon as reasonably practicable.

7. DRS'S OBLIGATIONS

- 7.1 DRS shall provide DB Cargo with a Specification as soon as reasonably practicable after it becomes aware that it desires use of the facility, and / or Facility Service(s). DRS shall use its reasonable endeavours to ensure that any Specification provided in accordance with this Clause 7.1 conforms to the Facility Operating Constraints.
- 7.2 DRS shall notify DB Cargo promptly of any changes to a Specification and shall use its reasonable endeavours to ensure that any such changes conform to the Facility Operating Constraints.
- 7.3 DRS shall inform DB Cargo in advance of any Dangerous Goods which will or might be carried in any of the Specified Equipment comprised in the Train Services and of any known or suspected defects in any of the Specified Equipment or any unusual specification or characteristic of any Goods carried in any of the Specified Equipment which might materially affect the operation of the Train Services or the facility or adversely affect the facility.
- 7.4 DRS shall operate the Train Services and manage any activity it carries out in connection with the Train Services at all times in such a way as to:
 - (a) minimise, by taking all reasonable steps, any nuisance to or disturbance of DB Cargo or any other person on the Premises or the owners or occupiers of other land whether within or outside the Premises; and
 - (b) prevent the escape of any Goods or other items or substances within the possession or control of DRS from the Specified Equipment.

8. GOODS

8.1 Risk

Risk in the Goods (including, without limitation, all risk of theft, loss or damage to the Goods) shall at all times remain with DRS and DB Cargo shall not be under any obligation to insure the Goods or any part thereof and shall have no liability in respect of the same unless and to the extent DB Cargo is in breach of its obligations under this contract.

8.2 Dangerous Goods

- 8.2.1 Dangerous Goods shall not be brought onto the facility or dealt with at any Terminal unless and until a declaration by DRS in writing shall have been given to DB Cargo of their nature and quantity, including details of special handling requirements in the event of an emergency, and DB Cargo's consent shall have been first obtained, and then only upon such terms and conditions as DB Cargo acting reasonably may require, or as may be prescribed in DB Cargo's bye-laws or by or under statute.
- 8.2.2 Subject to Clause 13, DB Cargo shall be permitted by DRS to give any particulars furnished under Clause 8.2.1 to whomsoever DB Cargo reasonably believes necessary either as a matter of law or for health and safety reasons.

8.3 Ability to reject Dangerous Goods

DB Cargo reserves the right to refuse to accept Dangerous Goods onto the facility where it reasonably believes that they may cause damage to the Premises, the facility, the environment, harm to human health or otherwise pose a health and safety risk.

9. ACCESS CHARGES

9.1 Obligation on DRS to pay

In respect of each Charging Period, DRS shall pay the Charges.

9.2 Price variation

- 9.2.1 The Access Sum and the Ancillary Charges shall remain in force until the day preceding 1 April 2017.
- 9.2.2 On 1 April in each Contract Year, commencing 1 April 2017, the Access Sum and the Ancillary Charges shall be adjusted by multiplying each of them by the Adjustment Factor (rounded to three decimal places) which shall have been calculated in accordance with the following formula:

Adjustment Factor =
$$1 + \frac{(RPI_{t-1} - RPI_{t-2})}{RPI_{t-2}}$$

where:

 RPI_{t-1} means the average value of the monthly figures of the General Index of Retail Prices All Items measured by CHAW ("**RPI**") for the 12 months up to and including the month of September immediately preceding the relevant 1 April; and

 RPI_{t-2} means the average value of the monthly figures of RPI for the 12 months up to and including the month of September which is 12 months before the relevant 1 January.

9.2.3 If RPI for any month of September shall not have been published, or there is a material change in the base composition of RPI, then the parties may agree to such other index as they deem appropriate with the object of placing both parties in the position in which they would have been had there been no change in the base composition of RPI.

- 9.2.4 The Access Sum and the Ancillary Charges for each forthcoming Contract Year may be reviewed by the parties if there is a Material Change in:
 - (a) the cost to DB Cargo of fulfilling its obligations under this contract; or
 - (b) the total number of train services permitted to use the facility or relevant parts of the facility.

On or before 1 December in each Contract Year, DB Cargo shall give notice that either it wishes to review the Access Sum and the Ancillary Charges, or that no such review is required. For the avoidance of doubt, a revision under this clause 9.2 may not be made more than once in any Contract Year.

- 9.2.5 For the purposes of reviewing the Access Sum and the Ancillary Charges:
 - (a) a **"Material Change"** shall be defined as a forecast increase or decrease of 10 percent or more in the:
 - (i) costs of operation of the facility or relevant Ancillary Services for the relevant year t+1 compared to the costs for the then current year t; or
 - (ii) the total number of train services for all users accessing the facility:

(after taking into account the Adjustment Factor) where "relevant year t+1" means the Contract Year for the purposes of which any calculation falls to be made and "relevant year t" means the then current Contract Year preceding the relevant year t+1; and

- (b) the Access Sum for the relevant year t+1 equals the forecast cost to DB Cargo of fulfilling its obligations under this contract for the relevant year t+1 divided by the forecast number of trains for the relevant year t+1.
- 9.2.6 Following a notice being given in accordance with Clause 9.2.4 above, DB Cargo shall provide such evidence in support of the material change or in support of its contention that there has been no material change in the cost (as the case may be) as DRS may reasonably request.
- 9.2.7 The figure(s) agreed by the parties for the purpose of Clauses 9.2.2 and 9.2.4 shall replace those set out in Schedule 5 on and from the start of the relevant year t+1 (which, for the avoidance of doubt, shall not be before 1 April of the relevant year) and shall be subject to a minimum notice period of 3 months.
- 9.2.8 If the parties fail to agree Access Sum and Ancillary Charges figure(s) for the relevant year t+1 within 28 days of a notice being given in accordance with Clause 9.2.4 above, either party shall be entitled to refer the determination of the Access Sum and Ancillary Charges either to arbitration in accordance with the Access Dispute Resolution Rules or to ORR for determination. DB Cargo shall notify ORR of the revised figure(s) within 28 days of them being agreed or determined.

9.3 Unpaid sums

If either party fails to pay:

- (a) any invoice issued to it under this contract in respect of the Charges; or
- (b) any sum which has fallen due in accordance with any other provision of this contract,

then, subject to Clause 15.1.1:

- (i) the amount invoiced or sum due, as referred to in Clause 9.3(a) or (b), shall immediately constitute a debt due and owing from the party who has failed to pay the invoice or sum due to the other party (and to any assignee of a party's right to payment in respect of any Charges or other sum due);
- (ii) such debt shall be recoverable by any means available under the laws of England and Wales; and
- (iii) unless both parties to this contract agree otherwise, the dispute resolution procedures in Clause 12 shall not apply to proceedings commenced under this Clause 9.3.

10 INDEMNITIES AND LIMITATION OF LIABILITY

10.1 DB Cargo indemnity

Subject to clause 10.4 DB Cargo shall (on an after tax basis) indemnify DRS, and keep it indemnified, against all damage, losses, claims, proceedings, demands, liabilities, costs, damages, orders and out of pocket expenses (including costs reasonably incurred in investigating or defending any claim, proceedings, demand or order and any expenses reasonably incurred in preventing, avoiding or mitigating loss, liability or damage) incurred or suffered by it as a result of any breach by DB Cargo of any of its obligations under this contract.

10.2 DRS indemnity

Subject to clause 10.4 DRS shall (on an after tax basis) indemnify DB Cargo, and keep it indemnified, against all damage, losses, claims, proceedings, demands, liabilities, costs, damages, orders and out of pocket expenses (including costs reasonably incurred in investigating or defending any claim, proceedings, demand or order and any expenses reasonably incurred in preventing, avoiding or mitigating loss, liability or damage) incurred or suffered by it as a result of any breach by DRS of any of its obligations under this contract.

10.3 Mitigation

A party wishing to claim under any indemnity provided for in this contract shall take all reasonable steps to prevent, mitigate and restrict the circumstances giving rise to that claim and any losses connected with that claim.

10.4 Notification of claims, limitation on liability and restriction on claims

10.4.1 Notification of claims and limitation on liability

Save as otherwise expressly provided in this contract, neither party shall be liable in

respect of any breach of this contract:

- a) unless notice of such breach is given by or on behalf of the claimant to the respondent, setting out detailed particulars of the grounds on which the relevant claim is based, within six months after the facts giving rise to the claim first became known by the claimant or could, with reasonable diligence, have become so known; and
- b) arising from any single occurrence or circumstance (or connected series of occurrences or circumstances), if the amount of the relevant claim exceeds the Liability Cap, to the extent it so exceeds this amount.

10.4.2 Liability Cap

The Liability Cap for the first Contract Year shall mean the sum of £1,000,000 and in relation to any subsequent Contract Year, the sum calculated in accordance with the following formula:

$$\mathbf{C}_{\mathbf{n}} = \mathbf{C}_{1} \quad \left[\begin{array}{c} \frac{RPI_{n}}{RPI_{I}} \end{array} \right]$$

where:

- (i) C_1 is the sum of £1,000,000;
- (ii) C_n is the Liability Cap in the nth subsequent Contract Year;
- (iii) RPI_n is the Retail Prices Index (defined as RPI in Clause 9.2.2) published or determined with respect to the first month of the subsequent Contract Year n; and
- (iv) RPI_I is the Retail Prices Index (defined as RPI in Clause 9.2.2) published or determined with respect to the first month of the first Contract Year.

10.4.3 Applicable Contract Year for indexation

For the purposes of determining the Liability Cap under Clause 10.4.2, the applicable Contract Year shall be the Contract Year in which the relevant incident, event or circumstance begins to occur.

- 10.4.4 *Exclusion of Environmental Damage* Not used.
- 10.4.5 Restriction on claims

Save as otherwise expressly provided in this contract, neither party may recover or seek to recover from the other party any amount in respect of any loss of revenue, loss of profits, goodwill, reputation or other consequential, indirect or special damages in connection with the subject matter of this contract, which is or is alleged to be caused to it by the other party, save in respect of death or injury to persons or physical damage to property.

11 GOVERNING LAW

This contract shall be governed by and construed in accordance with the laws of England and Wales.

12 DISPUTE RESOLUTION

- 12.1 All disputes or matters in difference between the parties of whatever nature arising out of or in connection with this contract ("**Disputes**") shall, except when otherwise expressly provided in this contract, be resolved in accordance with the terms of this Clause 12.
- 12.2 Any Dispute shall be notified by either party to the other promptly and then, in the first instance, be the subject of an extraordinary meeting between the parties. Such extraordinary meeting shall be held as soon as practicable after notification of a Dispute (and, in any event, within seven days of such notification being received). Each party agrees to procure that an authorised representative shall attend an extraordinary meeting called in accordance with this Clause 12.2, and those representatives of the parties attending the extraordinary meeting shall use all reasonable endeavours to resolve the Dispute.
- 12.3 If the Dispute cannot be resolved at the extraordinary meeting, it shall be referred by the parties to a person of director level at each of DB Cargo and DRS who shall cooperate in good faith to resolve the Dispute as amicably as possible within seven days of its referral (or such longer period as the parties may agree).
- 12.4 Any Disputes not resolved under Clause 12.3 may be referred by either party to arbitration in accordance with the Access Dispute Resolution Rules or may, if appropriate, be referred by either party to ORR for determination.
- 12.5 Each party shall bear its own costs in relation to the proceedings described in Clauses 12.2 and 12.3.

13. CONFIDENTIALITY

13.1 Confidential Information

13.1.1 General obligation

Except as permitted by Clause 13.2, all Confidential Information shall be held confidential during and after the continuance of this contract and shall not be divulged in any way to any third party without the prior written approval of the other party.

13.1.2 Affiliates

Except as permitted by Clause 13.2, each party shall procure that its Affiliates and its and their respective officers, employees and agents shall keep confidential and not disclose to any person any Confidential Information.

13.2 Entitlement to divulge

Either party, and its Affiliates, and its and their respective officers, employees and agents, shall be entitled in good faith to divulge any Confidential Information without the approval of the other party in the following circumstances:

- (a) to ORR;
- (b) to the Scottish Ministers;
- (c) to any Affiliate of either party;
- (d) to any officer or employee of the party in question or any person engaged in the provision of goods or services to or for him if disclosure is necessary or reasonably required to enable the party in question to perform its obligations under this contract, upon obtaining an undertaking of strict confidentiality from such officer, employee or person;
- (e) to any professional advisers or consultants of such party engaged by or on behalf of such party and acting in that capacity, upon obtaining an undertaking of strict confidentiality from such advisers or consultants;
- (f) to any insurer or insurance broker from whom such party is seeking insurance or in connection with the making of any claim under any policy of insurance, upon obtaining an undertaking of strict confidentiality from the insurer or insurance broker;
- (g) to any lender, security trustee, bank or other institution from whom such party is seeking or obtaining finance or credit support for such finance, or any advisers to any such entity, or any rating agency from whom such party is seeking a rating in connection with such finance or credit support, upon obtaining an undertaking of strict confidentiality from the entity, advisers or rating agency in question;
- (h) to the extent strictly required by the Act, any licence under section 8 of the Act or regulation 6 of the Railway (Licensing of Railway Undertakings) Regulations 2005 held by the party in question, any other applicable law, the rules of any recognised stock exchange or regulatory body;
- (i) to the extent that it has become available to the public other than as a result of a breach of confidence; or
- (j) under the order of any court or tribunal of competent jurisdiction.

13.3 Return of Confidential Information

Each of DB Cargo and DRS shall promptly return to the other party any Confidential Information requested by the other party if such request:

- (a) is made on or within two months after the Expiry Date or, if this contract is terminated earlier, is made within two months after the date on which this contract is terminated;
- (b) is reasonable; and
- (c) contains a sufficient description of the relevant Confidential Information to enable such information to be readily identified and located.

13.4 Retention or destruction of Confidential Information

If DB Cargo or DRS, as the case may be, has not received a request to return any Confidential Information to the other party under and within the time limits specified in Clause 13.3, it may destroy or retain such Confidential Information provided that such information shall remain subject to the confidentiality obligations in Clause 13.2.

13.5 Ownership of Confidential Information

All Confidential Information shall be and shall remain the property of the party which supplied it to the other party.

14. ASSIGNMENT AND SUB-CONTRACTING

14.1 Assignment

Neither party may assign, transfer, novate or create any encumbrance or other security interest over the whole or any part of its rights and obligations under this contract except to the extent approved by ORR following consultation with the other party, and subject to the conditions (if any) of ORR's approval.

14.2 Sub-contracting

- 14.2.1 Either party may sub-contract the performance of any of its obligations under this contract with the prior written agreement of the other party (such agreement not to be unreasonably withheld or delayed) without thereby relieving it of any such obligations to the other party.
- 14.2.2 Where a party has sub-contracted its rights or obligations under this contract to any third party in accordance with Clause 14.2.1, references to that party in this contract shall, with the exception of Clause 10 and without prejudice to Clause 14.2.1, include references to any sub-contractor so appointed.

15. PAYMENTS, INTEREST AND VAT

15.1 Payment

15.1.1 No deduction

All amounts due or payable by either party under this contract shall be paid free and clear of any deduction, withholding or set off, except:

- (a) as may be required by law; or
- (b) as expressly provided in this contract.

15.1.2 Delivery of invoices

All invoices, or statements of amounts payable, issued under any provision of this contract shall be delivered by hand at, or sent by prepaid first class post or by facsimile transmission (with confirmation copy by prepaid first class post) or by e-mail to, the appropriate address for service for the recipient specified in Schedule 1 and shall be deemed to have been received by the addressee in accordance with Clause 17.4.3.

- 15.1.3 *Payment and content of invoices and other statements of amounts payable* Each invoice and statement of amounts payable shall, unless otherwise stated in this contract:
 - (a) be paid within 28 days of the date of its receipt; and
 - (b) contain such detail as to the constituent elements of the amounts stated to be payable as shall be necessary so as to enable the person to whom it is given to understand and check it.

15.1.4 Method of payment

All payments shall be made by direct debit mandate or standing order mandate, CHAPS transfer, BACS transfer or other electronic or telegraphic transfer to a London clearing bank or such other financial institution as may be approved by the party entitled to the payment, such approval not to be unreasonably withheld or delayed.

15.1.5 Credit notes

Where a credit note has been issued in accordance with any provision of this contract, the party in receipt of the credit note shall be entitled to apply the amount specified in it against any amount payable by it under this contract or any future invoice or statement of amounts payable it may receive under this contract.

15.2 Disputed amounts

Except as otherwise provided in this contract, within 21 days of receipt of an invoice or statement of amounts payable issued under any provision of this contract, the recipient shall notify the issuer of any aspects of the invoice or statement which it disputes, giving reasons for any dispute, and shall pay the undisputed amount in accordance with the terms of the invoice. Except to the extent that disputes are so notified, the recipient shall be deemed to have agreed the contents of the invoice or statement.

15.3 Interest

Without prejudice to any other rights or remedies which one party may have in respect of the failure of the other party to pay any amount on the due date, amounts payable under this contract and not paid by the due date shall carry interest (to accrue daily and to be compounded monthly) at the Default Interest Rate from the due date until the date of actual payment (as well after judgment as before), except to the extent that late payment arises from any failure by the invoicing party to comply with Clause 15.1.2 or Clause 15.1.3(b).

15.4 VAT

15.4.1 Payment of VAT

Where any taxable supply for VAT purposes is made under or in connection with this contract by one party to the other the payer shall, in addition to any payment required for that supply, pay such VAT as is chargeable in respect of it.

15.4.2 Reimbursement of VAT

Where under this contract one party is to reimburse or indemnify the other in respect of any payment made or cost incurred by the other, the first party shall also reimburse any VAT paid by the other which forms part of its payment made or cost incurred to the extent such VAT is not available for credit for the other party (or for any person with whom the indemnified party is treated as a member of a group for VAT purposes) under sections 25 and 26 of the Value Added Tax Act 1994.

15.4.3 VAT credit note to be issued on repayment

Where under this contract any rebate or repayment of any amount is payable by one party to the other, and the first party is entitled as a matter of law or of HM Revenue and Customs practice to issue a valid VAT credit note, such rebate or repayment shall be paid together with an amount representing the VAT paid on that part of the consideration in respect of which the rebate or repayment is made, and the first party shall issue an appropriate VAT credit note to the other party.

16. FORCE MAJEURE EVENTS

- 16.1 A "**Force Majeure Event**" shall be deemed to occur if and to the extent that there occurs, at the facility or affecting the facility, any event or circumstance or any combination of events or circumstances beyond the reasonable control of either party to this contract which is either unforeseeable or, if foreseeable, could not have been avoided by any reasonable means, and which prevents, materially impedes or materially delays that party from performing any of its obligations under this contract.
- 16.2 Without prejudice to the generality of Clause 16.1, a "**Force Majeure Event**" shall include the following events (and any circumstances arising as a direct consequence of any of the following events):
 - (a) an act of the public enemy or terrorists or war (declared or undeclared), threat of war, revolution, riot, insurrection, civil commotion, demonstration or sabotage;
 - (b) acts of theft, vandalism, wilful damage or accidental damage or destruction of machinery, equipment, track or other infrastructure;
 - (c) natural disasters or phenomena, including extreme weather or environmental conditions (such as lightning, earthquake, hurricane, storm, fire, flood, drought or accumulation of snow or ice);
 - (d) nuclear, chemical or biological contamination;
 - (e) pressure waves caused by devices travelling at supersonic speeds;
 - (f) discovery of fossils, antiquities or unexploded bombs;
 - (g) strike or other industrial action which is a single circumstance and which also is a strike or industrial action in sectors of the economy other than the railway industry;
 - (h) acts, restrictions, bye-laws, prohibitions or measures of any kind on the part of any governmental, parliamentary or local authority; and

- (i) any act of DB Cargo, its servants or agents which, though deliberate, is reasonably necessary for the safety or preservation of persons, the Premises, the facility, Terminals or Goods.
- 16.3 Neither party to this contract shall be responsible for any failure to fulfil an obligation under this contract if and to the extent that such failure shall be caused by, or directly or indirectly is by reason of, a Force Majeure Event which makes it impossible or impracticable for that party to comply with such obligation, and if the affected party complies with Clauses 16.4 to 16.6 below.
- 16.4 A party affected by a Force Majeure Event shall promptly upon becoming aware of the occurrence of a Force Majeure Event use all reasonable endeavours to:
 - (a) minimise, and where practicable avoid, the effects of the Force Majeure Event on its ability to perform its obligations under this contract; and
 - (b) minimise the duration of the Force Majeure Event.
- 16.5 As soon as reasonably practicable after the commencement of the Force Majeure Event (and, in any event, within 72 hours of becoming aware of it), the affected party shall notify the other party of the Force Majeure Event and its consequences, the effects of the Force Majeure Event on its ability to perform any obligation(s) under this contract, the likely duration of such consequences and effects, and the remedial measures it proposes to avoid or remove the Force Majeure Event or to mitigate its consequences and effects.
- 16.6 The affected party shall promptly give the other party all other information concerning the Force Majeure Event and the steps which could reasonably be taken, and which the affected party proposes to take, to avoid or remove the Force Majeure Event or to mitigate its consequences and effects as may reasonably be requested by the other party from time to time, and promptly inform the other party of any material developments in relation to the Force Majeure Event.
- 16.7 The right of a party affected by a Force Majeure Event to relief under Clause 16.3 shall cease on the earlier of:
 - (a) the date on which its performance of any obligation(s) under this contract is no longer prevented, materially impeded or materially delayed by the Force Majeure Event; and
 - (b) the date on which such performance would no longer have been prevented, materially impeded or materially delayed if that party had complied with its obligations under Clause 16.4.
- 16.8 Neither party to this contract shall be entitled to claim that a Force Majeure Event has prevented it from paying any monies which it would otherwise be liable to pay under this contract.

17. MISCELLANEOUS

17.1 Non waiver

17.1.1 No waiver

No waiver by either party of any failure by the other to perform any obligation under this contract shall operate or be construed as a waiver of any other or further default, whether of a like or different character.

17.1.2 Failure or delay in exercising a right or remedy

The failure to exercise or delay in exercising a right or remedy under this contract shall not constitute a waiver of the right or remedy or a waiver of any other rights or remedies, and no single or partial exercise of any right or remedy under this contract shall prevent any further exercise of the right or remedy or the exercise of any other right or remedy.

17.2 Variations

- 17.2.1 *Amendments to be in writing and to be approved* No amendment of any provision of this contract shall be effective unless:
 - (a) such amendment is in writing;
 - (b) such amendment is signed by, or on behalf of, the parties; and
 - (c) if it is an amendment which:
 - (i) requires ORR's approval under section 22 of the Act; or
 - (ii) is made under section 22A or 22C of the Act or Schedule 4A to the Act,

the amendment has been so approved or directed by ORR as applicable.

17.2.2 *Exception to Clause 17.2.1(c)*

Clause 17.2.1(c) does not apply to amendments of the following kinds:

- (a) an amendment made by virtue of a general approval issued by ORR under section 22 of the Act; and
- (b) a modification made by virtue of Clause 17.4.2.

17.2.3 Conformed copy of contract

DB Cargo shall produce and send to DRS and to ORR a conformed copy of this contract within 28 days of the making of any amendment or modification to it (not including amendments to the Access Sum and the Ancillary Charges in accordance with Clause 9.2).

17.3 Entire contract and exclusive remedies

17.3.1 *Entire contract* Subject to Clause 17.3.3:

- (a) this contract contains the entire agreement between the parties in relation to the subject matter of this contract;
- (b) each party acknowledges that it has not been induced to enter into this contract in reliance upon, nor has it been given, any warranty, representation, statement, agreement or undertaking of any nature whatsoever other than as expressly set out in this contract and, to the extent that this is not the case, the relevant party unconditionally and irrevocably waives any claims, rights or remedies which it might otherwise have had in relation to any such matter; and
- (c) neither party shall have any right to rescind or terminate this contract either for breach of contract or for misrepresentation or otherwise, except as expressly provided for in this contract.

17.3.2 *Exclusive remedies*

Subject to Clause 17.3.3 and except as expressly provided in this contract:

- (a) neither party shall have any liability (including liability arising as a result of any negligence, breach of contract or breach of statutory obligation) to the other in connection with the subject matter of this contract; and
- (b) the remedies provided for in this contract shall be the sole remedies available to the parties in respect of any matters for which such remedies are available.

17.3.3 Fraud, death and personal injury

Without prejudice to the generality of this Clause 17.3, nothing in this contract shall exclude, restrict or limit, or purport to exclude, restrict or limit:

- (a) any liability which either party would otherwise have to the other party, or any right which either party may have to rescind this contract, in respect of any statement made fraudulently by the other party before the execution of this contract;
- (b) any right which either party may have in respect of fraudulent concealment by the other party;
- (c) any right which either party may have in respect of a statement of the kind referred to in section 146 of the Act, whether or not proceedings have been instituted in that respect; or
- (d) any liability which either party may have towards the other party for death or personal injury resulting from its negligence or the negligence of any of its officers, employees or agents.

17.4 Notices

17.4.1 Giving of notices

Any notice to be given under this contract:

(a) shall be in writing; and

(b) shall be duly given if signed by or on behalf of a person duly authorised to do so by the party giving the notice and delivered by hand at, or by sending it by prepaid first class post, or by facsimile transmission or e-mail (with confirmation copy by prepaid first class post) to, the relevant address or facsimile number or e-mail address set out in Schedule 1.

For the purposes of this Clause 17.4 and Clause 15.1.2, delivery by hand shall include delivery by a reputable firm of couriers.

17.4.2 Right to modify contact details

A party shall be entitled to modify in any respect the contact details which relate to it and which are set out in Schedule 1 by giving notice of such modification:

- (a) to the other party as soon as reasonably practicable; and
- (b) to ORR within 14 days of such modification.

17.4.3 Deemed receipt

A notice shall be deemed to have been given and received:

- (a) if sent by hand or recorded delivery, at the time of delivery;
- (b) if sent by prepaid first class post from and to any place within the United Kingdom, three Working Days after posting unless otherwise proven; or
- (c) if sent by facsimile (subject to confirmation of uninterrupted transmission by a transmission report) or by e-mail (subject to confirmation of opening the e-mail by return of a read receipt), before 1700 hours on a Working Day, on the day of transmission and, in any other case, at 0900 hours on the next following Working Day.

17.4.4 Copyees

If Schedule 1 specifies any person to whom copies of notices shall also be sent:

- (a) the party giving a notice in the manner required by this Clause 17.4 shall send a copy of the notice to such person at the address for sending copies as specified in Schedule 1, or to such other person or address as may, from time to time, have been notified by the party to be notified to the notifying party under this Clause 17.4; and
- (b) such copy notice shall be sent immediately after the original notice.

17.5 Counterparts

This contract may be executed in two counterparts which, taken together, shall constitute one and the same document. Either party may enter into this contract by signing either of such counterparts.

17.6 Survival

Those provisions of this contract which by their nature or implication are required to survive expiry or termination of this contract (including the provisions of Clauses 9.3 (Unpaid Sums), 10 (Indemnities and Limitation of Liability), 11 (Governing Law), 13

(Confidentiality), 15 (Payments, Interest and VAT), 16 (Force Majeure Events), paragraph 4 of Schedule 4 (Consequence of Termination)) shall so survive and continue in full force and effect, together with any other provisions of this contract necessary to give effect to such provisions.

17.7 Contracts (Rights of Third Parties) Act 1999

17.7.1 *Application to third parties*

Save as provided in this Clause 17.7, or as expressly provided elsewhere in this contract, no person who is not a party to this contract shall have any right under the Contracts (Rights of Third Parties) Act 1999 to enforce any term of this contract.

17.7.2 Application to ORR

ORR shall have the right under the Contracts (Rights of Third Parties) Act 1999 to enforce directly such rights as have been granted to it under this contract.

17.7.3 Contract amendments

Subject to Clause 17.2, DB Cargo and DRS shall not enter into any agreement with a third party that requires the consent of any third party in order to amend this contract.

IN WITNESS whereof this contract has been duly executed.

[SIGNED on behalf of DB Cargo International Limited 85

SIGNED on behalf of Direct Rail Services Limited

SCHEDULE 1 – CONTACT DETAILS

1. Addresses for Service

The parties shall use the following addresses for notification to DRS or DB Cargo, as appropriate, of the following issues:

1.1 For all matters excluding financial and operational management:

DB Cargo International Limited Lakeside Business Park Carolina Way Doncaster DN4 5PN

Attention: Company Secretary

Tel.: 01302 575157 FAX: 01302 575005

And copied to:

DB Cargo International Limited Lakeside Business Park Carolina Way Doncaster DN4 5PN

Attention: Head of Sector BIC

Tel.: 01302 575546 FAX: 01302 578233

And

Direct Rail Services Limited, Regents Court, Baron Way, Carlisle, CA6 4SJ.

Attention: Mr Barry Earl.

Tel No: 01228 406464.

1.2 For matters of a financial nature:

DB Cargo International Limited Lakeside Business Park Carolina Way Doncaster DN4 5PN

Attention: National Accounts Receivable

Tel.: 01302 575729

And

Direct Rail Services Limited, Regents Court, Baron Way, Carlisle, CA6 4SJ.

Attention: Accounts Payable.

Tel No: 01228 406634

1.3 For matters of a day-to-day operational nature:

DB Cargo International Limited Lakeside Business Park Carolina Road Doncaster DN4 5PN

Attention: Duty Control Manager Tel.: 01302 575047

And

Direct Rail Services Limited, Regents Court, Baron Way, Carlisle, CA6 4SJ.

Attention: Control Room.

Tel No: 01228 406619

SCHEDULE 2 – THE FACILITY, FACILITY OPERATING CONSTRAINTS, FACILITY SERVICES AND ANCILLARY SERVICES

1. The Facility

• The facility is the railway network within the Mossend Euroterminal, located adjacent to the Mossend Eurocentral complex near the town of Motherwell in the county of Lanarkshire of which DB Cargo is the facility owner.

2. The Plan

The Plan is appended to this Schedule 2.

3. Exchange Point

The relevant boundary between the Network and the facility as depicted on the plan.

4. **Opening Hours**

The facility is open 24hours Monday – Friday, closed between the hours of 12:00 - 16:00 Saturday and 04:00 - 18:00 Sunday.

5. Facility Operating Constraints

The Facility Operating Constraints will be specified in the relevant Safe System of Work, Standing Instructions and Special Instructions as amended from time to time, or otherwise published by DB Cargo from time to time.

The Facility has 6 lines, two of which can be used for Operational Stabling.

6. Facility Services

DB Cargo shall provide the following Facility Services to DRS:

- Access, including placement within and withdrawal from Terminal using DB Cargo groundstaff & class 08 shunt loco;
- arrival and departure train examination according to industry standards;
- Operational Stabling;

Provision of the Facility Services is included in the Access Sum set out in paragraph 1 of Schedule 5.

7. Ancillary Services

If requested, DB Cargo may provide the following Ancillary Services to DRS:

- loads examination, when required;
- additional shunting (using DRS or DB Cargo traction unit and/or groundstaff as appropriate and subject to availability);
- access for DRS maintenance teams to attend to vehicle defects;
- Stabling, subject to availability on the day;

Provision of these Ancillary Services will incur the Ancillary Charges set out in paragraph 2 of Schedule 5.

Additional ancillary services which are not subject to regulatory approval may be available and provided, subject to suitable separate contractual arrangements.





SCHEDULE 3 – THE TRAIN SERVICES

1. Train Services

The Train Services shall be listed and described by DRS in a Specification in the form appended to this Schedule 3.

2. Specified Equipment

2.1 The Specified Equipment - Passenger:

- Class 68 (or other classes of diesel locomotives operated by DRS and registered to run on the Network); and
- up to 6 x Mk.2 passenger vehicles per Train Service.

2.2 The Specified Equipment – Intermodal:

- Class 66 or 68 (or other classes of diesel locomotives operated by DRS and registered to run on the Network); and
- Up to 18 x IKA/IDA wagons conveying ISO containers or swap-bodies marshalled in trains with a maximum length of 656 metres / 103 SLU, typically comprising 36 platforms (Daventry trains) or 20 platforms (Inverness trains).

3. Input and information

DRS shall also provide the following input and information to DB Cargo:

- consultation during the development of the Safe System of Work;
- consultation during the development of Standing and Special Instructions;
- the provision of forward planning information for the purpose of compiling the Facility Workplan;
- attendance at facility operations meetings; and
- attendance of authorised personnel at specified incidents.

Appendix 1 – Form of Specification

Passenger

Train	Dep.	Days	Origin	Destination	Arr.	Su	Мо	Tu	We	Th	Fr	Sa	Comments
Arrival	-												
5Kxx	XX:XX	SuO	Xxxxxxxxx	Mossend ET	18:15	Х							Departs as 5K18 Monday
5Kxx	XX:XX	SuO	Xxxxxxxxx	Mossend ET	18:15	Х							Departs as 5K03 Monday
<u>5K14</u>	19:47	FSX	Edinburgh	Mossend ET	21:05		Х	Х	Х	Х			Departs as 5K18 Tuesday-Friday
<u>5L70</u>	18:24	FSX	Cardenden	Mossend ET	20:27		Х	Х	Х	Х			Departs as 5K03 Tuesday-Friday
Departure													
5K03	04:30	SX	Mossend ET	Edinburgh	-		Х	Х	Х	Х	Х		Formed off 5Kxx SuO / 5L70 FSX
5K18	04:35	SX	Mossend ET	Cardenden	-		Х	Х	Х	Х	Х		Formed off 5Kxx SuO / 5K14 FSX

Train Services are each formed of Class 68 plus up to 6 Mk.2 coaches, including DVT. Train Services will be stabled and unplanned maintenance may also be undertaken, subject to risk assessment and the approval of the DB Cargo responsible manager. Shunting may be undertaken utilising DRS' staff under the control and by agreement with the DB Cargo PIC, in accordance with the DB Cargo Safe System of Work.

Tittel modal												
Dep.	Days	Origin	Destination	Arr.	Su	Мо	Tu	We	Th	Fr	Sa	Comments
06:16	SX	Daventry IFT	Mossend ET	13:59		Х	Х	Х	Х	Х		Departs as 4M48 Monday-Friday
06:40	SO	Daventry IFT	Mossend ET	14:00							Х	Departs as 4M48 Sunday
13:19	SX	Inverness	Mossend ET	17:56		Х	Х	Х	Х	Х		Departs as 4H47 Tuesday-Saturday
13:19	SO	Inverness	Mossend ET	17:54							Х	Departs as 4H47 Monday
ıre												
05:04	SO	Mossend ET	Inverness								Х	Formed off 4D47 FO
05:05	SX	Mossend ET	Inverness			Х	Х	Х	Х	Х		Formed off 4D47 SO / FSX
14:43	SuO	Mossend ET	Daventry IFT	21:37	Х							Formed off 4S43 SO
19:04	SX	Mossend ET	Daventry IFT	02:02/02	:34	Х	Х	Х	Х	Х		Formed off 4S43 Monday-Friday
	06:16 06:40 13:19 13:19 13:19 05:04 05:05 14:43	06:16 SX 06:40 SO 13:19 SX 13:19 SO ure 05:04 SO 05:05 SX 14:43 SuO	06:16SXDaventry IFT06:40SODaventry IFT13:19SXInverness13:19SOInvernessure05:04SOMossend ET05:05SXMossend ET14:43SuOMossend ET	06:16SXDaventry IFTMossend ET06:40SODaventry IFTMossend ET13:19SXInvernessMossend ET13:19SOInvernessMossend ETure05:04SOMossend ET05:05SXMossend ETInverness14:43SuOMossend ETDaventry IFT	06:16SXDaventry IFTMossend ET13:5906:40SODaventry IFTMossend ET14:0013:19SXInvernessMossend ET17:5613:19SOInvernessMossend ET17:54ure05:04SOMossend ETInverness05:05SXMossend ETInverness.14:43SuOMossend ETDaventry IFT21:37	06:16SXDaventry IFTMossend ET13:5906:40SODaventry IFTMossend ET14:0013:19SXInvernessMossend ET17:5613:19SOInvernessMossend ET17:54ure05:04SOMossend ETInverness05:05SXMossend ETInverness.14:43SuOMossend ETDaventry IFT21:37	Dep.DaysOriginDestinationArr.SuMo06:16SXDaventry IFTMossend ET13:59X06:40SODaventry IFTMossend ET14:0013:19SXInvernessMossend ET17:56X13:19SOInvernessMossend ET17:54ure05:04SOMossend ETInverness.05:05SXMossend ETInverness.X14:43SuOMossend ETDaventry IFT21:37X	Dep.DaysOriginDestinationArr.SuMoTu06:16SXDaventry IFTMossend ET13:59XX06:40SODaventry IFTMossend ET14:00XX13:19SXInvernessMossend ET17:56XX13:19SOInvernessMossend ET17:54Xure05:04SOMossend ETInverness.X05:05SXMossend ETInverness.XX14:43SuOMossend ETDaventry IFT21:37XX	Dep.DaysOriginDestinationArr.SuMoTuWe06:16SXDaventry IFTMossend ET13:59XXXX06:40SODaventry IFTMossend ET14:00XXXX13:19SXInvernessMossend ET17:56XXXX13:19SOInvernessMossend ET17:54XXXure05:04SOMossend ETInverness.XXX14:43SuOMossend ETDaventry IFT21:37XXX	Dep.DaysOriginDestinationArr.SuMoTuWeTh06:16SXDaventry IFTMossend ET13:59XXXXX06:40SODaventry IFTMossend ET14:00XXXXX13:19SXInvernessMossend ET17:56XXXXX13:19SOInvernessMossend ET17:54XXXXure05:04SOMossend ETInverness.XXXX14:43SuOMossend ETDaventry IFT21:37XXXX	Dep.DaysOriginDestinationArr.SuMoTuWeThFr06:16SXDaventry IFTMossend ET13:59XXXXXX06:40SODaventry IFTMossend ET14:00XXXXXX13:19SXInvernessMossend ET17:56XXXXXX13:19SOInvernessMossend ET17:54ure05:04SOMossend ETInverness.XXXXX14:43SuOMossend ETDaventry IFT21:37X	Dep.DaysOriginDestinationArr.SuMoTuWeThFrSa06:16SXDaventry IFTMossend ET13:59XXXXXX06:40SODaventry IFTMossend ET14:00XXXXX13:19SXInvernessMossend ET17:56XXXXX13:19SOInvernessMossend ET17:54XXXXure05:04SOMossend ETInverness.XXXX05:05SXMossend ETInverness.XXXX14:43SuOMossend ETDaventry IFT21:37XXXXX

Intermodal

The Daventry wagon set comprises up to 18 IKA/IDA wagons and the Inverness set up to 10 IKA wagons. Train Services will be unloaded and reloaded using DB Cargo staff and equipment, subject to a separate contract. Train Services will be examined on arrival and prior to departure by DB Cargo groundstaff in accordance with industry standards, DRS' method statement and with the DB Cargo Safe System of Work. Shunting may be undertaken by DRS' or DB Cargo groundstaff, utilising DRS' or DB Cargo locomotive(s) according to availability, under the control of the DB Cargo PIC and in accordance with the DB Cargo Safe System of Work. DRS' Field Support Team will be granted access to attend to defective wagons and undertake limited regular maintenance, typically to change brake-blocks.

SCHEDULE 4 - EVENTS OF DEFAULT, SUSPENSION AND TERMINATION

1. Events of Default

1.1 DRS Events of Default

The following are DRS Events of Default:

- (a) DRS ceases to be authorised to be the operator of trains for the provision of the Train Services by a Non-Passenger Train Operator's Licence granted under section 8 of the Act or a European Freight Licence and a valid Statement of National Regulatory Provisions granted under regulations 6 and 10 respectively of the Railway (Licensing of Railway Undertakings) Regulations 2005, unless it is exempt from the requirement to be so authorised;
- (b) an Insolvency Event occurs in relation to DRS;
- (c) (i) any breach by DRS of this contract or its Safety Obligations; or
 - (ii) any event or circumstance which is reasonably likely to result in any such breach,

which, by itself or taken together with any other such breach, event or circumstance, DB Cargo reasonably considers constitutes a threat to the safe operation of any part of the facility or the Premises;

- (d) any Charges or other amount due by DRS to DB Cargo under this contract remain unpaid for more than seven days after their due date;
- (e) any breach of this contract by DRS which, by itself or taken together with any other such breach, results, or is likely to result, in material financial loss to DB Cargo; and
- (f) any breach of this contract by DRS which, by itself or taken together with any other such breach, results, or is likely to result, in material disruption to the operation of the facility or the Premises.

1.2 *Notification*

The Train Operator shall notify DB Cargo promptly on becoming aware of the occurrence of a DRS Event of Default.

1.3 DB Cargo Events of Default

The following are DB Cargo Events of Default:

(a) DB Cargo ceases to be authorised to be the operator of the facility by a Network Licence granted under section 8 of the Act, unless exempt from the requirement to be so authorised;

- (b) an Insolvency Event occurs in relation to DB Cargo;
- (c) (i) any breach by DB Cargo of this contract or its Safety Obligations; or
 - (ii) any event or circumstance which is reasonably likely to result in any such breach,

which, by itself or taken together with any other such breach, event or circumstance DRS reasonably considers constitutes a threat to the safe operation of the Train Services; and

(d) any breach of this contract by DB Cargo which, by itself or taken together with any other such breach, results, or is likely to result, in material financial loss to DRS.

1.4 *Notification*

DB Cargo shall notify DRS promptly on becoming aware of the occurrence of an DB Cargo Event of Default.

2. Suspension

2.1 *Right to suspend*

Either party may serve a Suspension Notice on the other if an Event of Default has occurred and is continuing, provided the relevant Event of Default is reasonably capable of remedy.

- 2.2 Contents of Suspension Notice A Suspension Notice shall specify:
 - (a) the nature of the relevant Event of Default;
 - (b) the date and time at which suspension is to take effect;
 - (c) in the case of a Suspension Notice served on DRS, reasonable restrictions imposed on the permission to use the facility or any parts of it while the Suspension Notice is in force;
 - (d) in the case of a Suspension Notice served on DB Cargo, details of any necessary suspension of the Train Services while the Suspension Notice is in force;
 - (e) the steps reasonably required to remedy the Event of Default; and
 - (f) a reasonable grace period for the defaulting party to remedy it (where the Event of Default is a failure by DRS to pay any part of the Charges or other amounts due under this contract, seven days shall be a reasonable grace period).
- 2.3 *Effect of a Suspension Notice served by DB Cargo* Where DB Cargo has served a Suspension Notice on DRS:

- (a) DRS shall comply with any reasonable restrictions imposed on it by the Suspension Notice;
- (b) the Suspension Notice shall remain in full force and effect in accordance with its terms until it has been revoked either in whole or in part by notice from DB Cargo to DRS under paragraph 2.5.4; and
- (c) service of the Suspension Notice shall not affect DRS's continuing obligation to pay the Charges.

2.4 *Effect of a Suspension Notice served by DRS* Where DRS has served a Suspension Notice on DB Cargo:

- (a) it shall have the effect of suspending DRS's permission to use the facility to operate the Train Services to the extent specified in the Suspension Notice; and
- (b) the Suspension Notice shall remain in full force and effect in accordance with its terms until it has been revoked either in whole or in part by notice from DRS to DB Cargo under paragraph 2.5.4.
- 2.5 *Suspension to be proportionate to breach*
- 2.5.1 A Suspension Notice served under paragraph 2.1 in respect of any of DRS Events of Default specified in paragraphs (a) and (c) to (f) (inclusive) of paragraph 1.1 shall, so far as reasonably practicable, apply only to the:
 - (a) Specified Equipment; and
 - (b) Train Services,

or part or parts of them, to which the relevant Train Operator Event of Default relates.

- 2.5.2 A Suspension Notice served under paragraph 2.1 in respect of any of the DB Cargo Events of Default specified in paragraphs 1.3 (a), (c) and (d) shall, so far as reasonably practicable, apply only to the part or parts of the facility to which the relevant DB Cargo Event of Default relates.
- 2.5.3 The party served with a Suspension Notice shall, with all reasonable diligence, take such steps as are specified in the Suspension Notice to remedy the Event of Default and keep the party which served the Suspension Notice fully informed of its progress.
- 2.5.4 Where a party served with a Suspension Notice has complied with its obligations under paragraph 2.5.3, whether in whole or in part, and it is reasonable for the suspension effected by the Suspension Notice to be revoked, whether in whole or in part, the party which served the Suspension Notice shall revoke the suspension to that extent. Such revocation shall be effected as soon as practicable after the remedy in question by notice to the other party specifying the extent of the revocation and the date on which it is to have effect.

3. Termination

3.1 *DB Cargo's right to terminate*

DB Cargo may serve a Termination Notice on DRS:

- (a) where, except during the period of a Suspension Notice relating to it, a DRS Event of Default has occurred and is continuing;
- (b) where DRS fails to comply with any material restriction in a Suspension Notice provided that the relevant DRS Event of Default is continuing;
- (c) where DRS fails to comply with its obligations under paragraph 2.5.3, provided that the relevant DRS Event of Default is continuing; or
- (d) where DB Cargo is serving three months' notice of its intention to terminate.

3.2 DRS's right to terminate

DRS may serve a Termination Notice on DB Cargo:

- (a) where, except during the period of a Suspension Notice relating to it, an DB Cargo Event of Default has occurred and is continuing;
- (b) where DB Cargo fails to comply with its obligations under paragraph 2.5.3, provided that the relevant DB Cargo Event of Default is continuing; or
- (c) where DRS is serving three months' notice of its intention to terminate.

3.3 *Contents of Termination Notice*

A Termination Notice shall specify:

- (a) the nature of the relevant Event of Default or other matter entitling termination under paragraphs 3.1 or 3.2 as the case may be;
- (b) a date and time, which shall be reasonable in the circumstances, at which termination is to take effect; and
- (c) where the Event of Default is capable of remedy:
 - (i) the steps reasonably required to remedy the Event of Default; and
 - (ii) a reasonable grace period within which such steps may be taken (and where the Event of Default is a failure by DRS to pay any part of the Charges or other amounts due under this contract, seven days shall be a reasonable grace period).

3.4 *Effect of Termination Notice*

Where DB Cargo or DRS has served a Termination Notice on the other:

- (a) the service of the Termination Notice shall not affect the parties' continuing obligations under this contract up to the date of termination, which date shall be determined in accordance with paragraph 3.4(c);
- (b) the party which has served the Termination Notice shall withdraw it by notice to the other party, upon being reasonably satisfied that the relevant Event of Default has been remedied; and
- (c) this contract shall terminate on the later of:
 - (i) the date and time specified in the Termination Notice for the contract to terminate (or such later date and time as the party which served the Termination Notice notifies to the other before the date and time so specified); and
 - (ii) the date on which a copy of the Termination Notice is given to ORR.

4. Consequence of termination

4.1 *Directions regarding location of Specified Equipment etc*

Immediately before, upon or following termination or expiry of this contract, DRS shall comply or procure compliance with all reasonable directions given by DB Cargo concerning the location of the Specified Equipment and other things left on the facility.

4.2 *Failure to comply with directions*

If DRS fails to comply with any directions given under paragraph 4.1, DB Cargo shall be entitled to remove from the facility any Specified Equipment and other things left on the facility or to instruct a third party to do so and any reasonable costs incurred by DB Cargo in taking such steps shall be paid promptly by DRS.

4.3 *Evidence of costs*

DB Cargo shall provide such evidence of such costs as are referred to in paragraph 4.2 as DRS shall reasonably request.

SCHEDULE 5 – CHARGES

1. Access Sum

The Access Sum is the sum of the charges for the Facility Services identified in paragraph 6 of Schedule 2, as follows:

For Passenger Train Services:

- 1.1 Access (per Train Service arrival plus departure):
- 1.2 Operational Stabling (per Train Service):
- 1.3 Other:

For Intermodal Train Services:

- 1.4 Access (per Train Service arrival plus departure):
- 1.5 arrival and departure, train examination:
- 1.6 Operational Stabling: per vehicle per day

For all train services not included in a Specification (accepted in accordance with clause 6.2(b) – the Premium Rates

- 1.7 Access no prior advice:
- 1.8 Access less than 24hrs notice via company "control" channels:

2. Ancillary Charges

The Ancillary Charges for provision by DB Cargo of the Ancillary Services specified in paragraph 7 of Schedule 2 are as follows:

- 2.1 Shunting, per hour (using DRS traction unit / DB Cargo groundstaff):
- 2.2 Shunting, per hour (using DB Cargo traction unit and groundstaff):

2.3 Stabling: per vehicle per day – first 24hrs free, then:

2.4 n/a

included in Access

n/a



